Author’s response to reviews

Title: PHYTOCHEMICAL ANALYSIS, ANTIOXIDANT AND ANTI-INFLAMMATORY POTENTIAL OF FERETIA APODANTHERA ROOT BARK EXTRACTS

Authors:

Oluwayinka Owolabi (drealyinks@yahoo.com)

Dorcas James (dbjams16187@yahoo.com)

Sani Ibrahim (sanibro2001@yahoo.com)

Binda Andongma (bindaandongma@yahoo.com)

Barnabas Kure (profbtk@gmail.com)

Version: 3 Date: 14 Oct 2017

Author’s response to reviews:

Owolabi Oluwayinka

Bioresources Development Centre

National Biotechnology Development Agency

Musa Ya’adua way, Lugbe

Abuja, FCT, Nigeria

14th October 2017

Editor-in-Chief

BMC Complementary & Alternative Medicine

Dear Reviewers,
We are grateful for the prompt and thorough review of the original manuscript with the topic “phytochemical analysis, antioxidant and anti-inflammatory potential of Feretia apodanthera root bark extracts” by Owolabi O.O., James, D.B., Ibrahim, S., Andongma, B.T., Fasanya O.O. and Kure, B. which was submitted for publication consideration in the BMC Complementary & Alternative Medicine. We appreciate the effectiveness of every reviewer in noting the actions needed to be taken to improve this manuscript and we are grateful for informing us about the publication eligibility of our manuscript in your highly regarded journal. We are writing in response to the remarks made by the reviewers.

In response to the editor’s comment, we are glad to say that the GC-MS analysis has been done and we hope that it has increased the value of this manuscript as suggested.

I will like to bring to your attention the addition of one author, Mr. Opeoluwa Fasanya who helped with the GCMS and FTIR analysis aspect of this research. We hope that our request be granted us and the name included.

We wish to thank Wissam Faour, Ph.D. (Reviewer 1) for his effective review on our manuscript. We are glad we were able to respond adequately.

In response to the reports by Chandrashekar Shreedhara (Reviewer 3), the conclusion has been edited to suit this manuscript better.

Thank you for your consideration, we look forward to your favourable response.

Kind regards,

Owolabi Oluwayinka
Scientific Officer
Bioresources Development Centre
National Biotechnology Development Agency